Author's response to reviews

Title: Diagnostic accuracy of cerebrospinal fluid protein markers for sporadic Creutzfeldt-Jakob disease in Canada: a 6-year prospective study

Authors:

Michael B Coulthart (michael.coulthart@phac-aspc.gc.ca)
Gerard H Jansen (gjansen@ottawahospital.on.ca)
Elina Olsen (elina.olsen@phac-aspc.gc.ca)
Debra L Godal (debra.godal@phac-aspc.gc.ca)
Tim Connolly (tim.connolly@phac-aspc.gc.ca)
Bernard CK Choi (bernard.choi@phac-aspc.gc.ca)
Zheng Wang (zheng.wang@phac-apsc.gc.ca)
Neil R Cashman (neil.cashman@vch.ca)

Version: 7 Date: 25 October 2011

Author's response to reviews:

Dear BMC Series Editorial Production Team,

Thank you kindly for your e-mail of October 25, 2011, in which you specify the remaining formatting changes required for our manuscript, MS ID 5443643715747675. Please accept my thanks for your careful editing, and my apologies for the residual problems. Here are details of the changes:

Major revisions:

1. Reference numbers have been corrected, with reference 39 (which was no longer needed in the text after the last revision of the MS) deleted from the reference list and the numbering of all subsequent references in the reference list corrected accordingly in both the list and the text. In addition, the field codes inserted by the citation-manager program (RefWorks) have now been removed, to leave only text-based information in the document.

Minor revisions:

2. The reference list has been moved to precede the tables.

Best regards,

Michael B. Coulthart, PhD